Author’s response to reviews

Title: Case Report: Severe Reversible Cardiomyopathy Associated with Systemic Inflammatory Response Syndrome in the Setting of Diabetic Hyperosmolar Hyperglycemic Non-ketotic Syndrome

Authors:

Justin Berk (justin.berk@gmail.com)
Raymond Wade (rcw658@gmail.com)
Hatice Bas (H.Duygu.baser@ttuhsc.edu)
Joaquin Lado (joaquin.lado@ttuhsc.edu)

Version: 1 Date: 15 Aug 2015

Author’s response to reviews:

Thank you for the feedback - it is appropriate and well-received. Included in this revision is all available follow-up lab data which demonstrates no significant changes on EKG on Day 4 and resolution of the inflammatory state based on laboratory data. This is the best data available to address the concerns of the reviewer and provides some insight into the follow-up of the patient.

Thank you for your input and your time.